Author's response to reviews

Title: The impact of vitamin D supplementation on vascular stiffness and blood pressure in Chronic Kidney Disease patients

Authors:

Adeera Levin (ALEvin@providencehealth.bc.ca)
Taylor Perry (TPerry@providencehealth.bc.ca)
Prathibha De Zoysa (PDeZoysa@providencehealth.bc.ca)
Mhairi K Sigrist (msigrist@bcgsc.ca)
Karin Humphries (KHumphries@providencehealth.bc.ca)
Mila Tang (MxTang@providencehealth.bc.ca)
Ognjenka Djurdjev (ODjurdjev@phsa.ca)

Version: 4 Date: 9 October 2014

Author's response to reviews:

Dear Reviewers,

Thank you for your comments. We have now clarified the formulations for the study drug: Calcitriol and 25-hydroxyvitamin D3.

We hope that these edits would be satisfactory.

Regards,
Adeera Levin